Author's response to reviews

Title: 'Pragmatic randomized controlled trial of individually prescribed exercise versus usual care in a heterogeneous cancer survivor population': A feasibility study PEACH Trial: Prescribed exercise after chemotherapy

Authors:

Julie M Walsh (walshj7@tcd.ie)
Juliette Hussey (jmhussey@tcd.ie)
Emer Guinan (emguinan@tcd.ie)
Dearbhaile O' Donnell (dodonnell@stjames.ie)

Version: 5 Date: 8 February 2010

Author's response to reviews:

To Whom it concerns,

Please find all relevant changes attached.

1. The title in the submission also contains PEACH trail: Prescribed Exercise After Chemotherapy.

2. Affiliations for each author have been included.

3. The abstract in the submission and in the manuscript are now the same.

4. Unfortunately I cannot give a more accurate citation for reference 48. The main author, Julie Walsh, is unavailable at present and I am unaware of any more details of the manual in question. Apologies for this.

If you have any further queries please do not hesitate to email us.

Kind Regards
Emer Guinan (Co-Author)